Author’s response to reviews

Title: Optimal Blood Pressure Control for Patients after Thoracic Endovascular Aortic Repair of Type B Aortic Dissection

Authors:

Nan Lu (lunansd@163.com)
Xiaojing Ma (13592807059@163.com)
Tan Xu (582890886@qq.com)
Zhuoqiao He (lunanbaobei@126.com)
Bayi Xu (sdzt1@163.com)
Qingfeng Xiong (ophthal.tzhang@hotmail.com)
Xuerui Tan (xueruitan@163.com)

Version: 2 Date: 09 May 2019

Author’s response to reviews:

May. 10, 2019

To:

Editorial Office

BMC Cardiovascular Disorders

Dear editors,

Manuscript ID: BCAR-D-19-00127

Title: Optimal Blood Pressure Control for Patients after Thoracic Endovascular Aortic Repair of Type B Aortic Dissection

Authors: Nan Lu, MD; Xiaojing Ma, MD; Tan Xu, MD; Zhuoqiao He, MD; Bayi Xu, MD; Qingfeng Xiong, MD; xuerui tan

Journal: BMC Cardiovascular Disorders
Thank you very much for your e-mail. We would like to thank you for your thoughtful suggestions, which helped us tremendously in improving our manuscript. We have carefully revised the manuscript accordingly.

Comments from the editor:

1. We have noticed that there is a discrepancy in the email address of the corresponding author between the submission system (xueruitan@163.com) and the manuscript file (tanxuerui@vip.sina.com) - please correct so they are consistent with each other.

Answer: We appreciated your professional review work on our manuscript. The corresponding author have two email addresses. We have changed tanxuerui@vip.sina.com to xueruitan@163.com on the title page of the manuscript.

2. Please include the email addresses for all authors on the title page. The corresponding author should still be indicated. Please also ensure these email addresses match the email addresses provided in the editorial manager system.

Answer: Thanks. We have added the email addresses for all authors on the title page.

3. Please add a "List of abbreviations" section after the "Conclusions" section in the manuscript body.

Answer: Thanks. We have added the "List of abbreviations" section after the "Conclusions" section in the manuscript body.

4. Kindly provide Consent to publish in the Declaration section

If your manuscript contains any individual person’s data in any form (including individual details, images or videos), consent for publication must be obtained from that person, or in the case of children, their parent or legal guardian. If your manuscript does not contain individual patients' data, this statement should be 'Not applicable'.

Answer: Thank you for your kind comments on our manuscript. We have added the consent for publication in the Declaration section.
Thank you very much to the reviewers and editor again for your significant comments and valuable suggestions to improve the quality of our manuscript.

Please address all correspondence to:

Xuerui Tan M.D., Ph.D., Prof. of Cardiology,
Department of Cardiology, The First Affiliated Hospital,
Shantou University Medical College,
No. 57, Changping Road, Shantou, Guangdong 515041, China

E-mail: xueruitan@163.com,
Phone: +86-754-88905455,
Fax: +86-754-88611690,

Yours faithfully,

Xuerui Tan